Sustainability orientation, industrial big data and product innovation: evidence from the European manufacturing sector

Abstract

Sustainability is one of the greatest challenges for industry today. The purpose of this paper is to study the influence of sustainability orientation on product innovation in the European manufacturing sector, with a particular focus on the direct and mediating effect of industrial big data use, something that has been largely neglected so far. The data used for the purpose of the present study were collected from the European Manufacturing Survey (EMS) 2018 edition, consisting of 1,123 surveys administered in Austria, Spain, Croatia, Lithuania, Slovakia, Slovenia and Serbia. Binary logistic regressions and Hayes mediation models are used to test the hypotheses. Results suggest that sustainability orientation practices and industrial big data use positively influence product innovation, and that industrial big data use mediates the relation between sustainability orientation and product innovation. The findings have implications for both theory and practice.

Keywords: Sustainability orientation, sustainability practices, industrial big data, product innovation, European Manufacturing Survey.

Funding: Open Access funding provided thanks to the CRUE-CSIC agreement with Elsevier

Declarations of interest: none

1. Introduction

The declaration of the United Nations Sustainable Development Goals in September 2015 marked a pivotal moment in an era increasingly attuned to the imperatives surrounding sustainable development. These 17 goals, targeted for attainment by 2030, have gained attention and spurred concerted efforts across diverse sectors. Within the private sector, the sustainability focus resonates with the earlier concept of the 'triple bottom line', a holistic approach considering economic, social and environmental aspects (Elkington and Rowlands, 1999). However, despite the growing emphasis on sustainability in business strategies, its seamless integration, particularly in the context of product innovation, remains intricate and subject to debate.

In this regard, sustainability orientation (SO) in business, defined as the extent to which firms integrate sustainability principles into their culture and practices, has become increasingly relevant (Claudy et al., 2016; Khizar et al., 2021; Kuckertz & Wagner, 2010; Usman Khizar et al., 2022). However, the inherent tensions between the economic, social and environmental aims of sustainability often pose challenges, particularly when developing new products and services (Fisher et al., 2020). The academic discourse reflects this complexity, with studies exposing varying perspectives on the relationship ranging from a positive outlook, in the sense that sustainability enhances product and service innovation (Ahmadi-Gh & Bello-Pintado, 2021), to negative (Sen & Bhattacharya, 2001) and ambiguous (Abdul-Rashid et al., 2017; Surroca et al., 2010) ones.

In the manufacturing industry, which represents an important part of the private sector, embracing sustainability can enhance product quality, resource efficiency and working conditions, while reducing costs and environmental impacts (Moldavska & Welo, 2017). Sustainability has particularly been seen as an important aspect in the development of product innovation (Keskin et al., 2020), technological innovation and eventually competitive advantage (Ahmadi-Gh & Bello-Pintado, 2021; Claudy et al., 2016). Indeed, sustainability and product innovation are at the heart of the success of today's manufacturing businesses, so learning more about how they connect is critical and largely underexplored (Ahmadi-Gh and Bello-Pintado 2021; Hallstedt, Thompson, and Lindahl 2013; Lintukangas, Kähkönen, and Hallikas 2019). Furthermore, given the intricate and multidimensional nature of integrating sustainability objectives into business processes, scholars are calling for the construction of models that consider missing data and for the investigation of mediating mechanism factors (Ahmadi-Gh & Bello-Pintado, 2021; Akomea et al., 2022; Claudy et al., 2016; Roxas et al., 2017).

Sustainable development is inextricably linked to technological advancement, which itself is increasingly related to the use of big data (Bashtannyk et al., 2020; Huber, 2004; Yudhistyra et al., 2020b). Data generated by different means such as machine sensors, RFID and the Internet of things, together with the decreasing cost of data gathering, processing and storage, are facilitating massive data generation known as big data (Chatterjee et al., 2022). The advent of big data has heralded a new era in the manufacturing sector, particularly in Europe, where the fusion of traditional manufacturing practices with digital technology is reshaping the industry landscape (Zhong et al., 2016). Big data, characterised by its volume, velocity, variety, veracity and value-adding role (Addo-Tenkorang & Helo, 2016; Zhong et al., 2016), has become a pivotal element in this transformation. Industrial big data is a subset of big data sourced from various elements such as devices, control units, robots and other equipment on the factory floor (Kirmse et al., 2019). This research is focused on industrial big data use, understood as the application of big data collected in manufacturing settings to improve decision making within industrial settings.

Data-driven innovation is a crucial growth pillar and source of competitive advantage (Chatterjee et al., 2022). Big data is helping companies with detecting new opportunities, and is seen as the next milestone for innovation, competition and productivity (Ciampi et al., 2021; Hämäläinen & Inkinen, 2019). However, the effect of SO and big data use on product innovation has so far been little explored. While previous research has looked at the effect of other digital technologies on product innovation and the relation between SO and product innovation, the specific role of big data remains largely uncharted (Zhao et al., 2021).

To address these calls, this paper aims to answer the following questions: Do sustainability orientation practices and industrial big data use positively influence product innovation? Is there a

mediating effect of industrial big data use on the relationship between sustainability orientation and product innovation? To answer these questions, three hypotheses are tested using data from the European Manufacturing Survey (EMS) 2018 edition, consisting of 1,123 surveys of firms located in seven European countries.

This study makes several contributions to the literature. First, the study focuses on how internal sustainability practices influence product innovation, contributing to the research line on sustainability-driven product innovation (Ahmadi-Gh & Bello-Pintado, 2021; Claudy et al., 2016; Keskin et al., 2020; Obal et al., 2020; Zhao et al., 2021), with the originality of exploring the role of industrial big data use. This analysis is pivotal for advancing our understanding of how sustainability practices and industrial big data usage synergise, providing a novel insight into their complementary effects.

Second, unlike many qualitative studies in this area (Keskin, Wever, and Brezet 2020; Zhao et al. 2021; Hallstedt, Isaksson, and Öhrwall Rönnbäck 2020), the present research employs quantitative methods, drawing on a detailed questionaire and an international sample that integrates manufacturing firms from across different industries.

The research also ventures into uncharted territory by simultaneously examining the relationship between sustainability, technology and innovation, filling a relevant gap in the literature where the confluence of these factors has not been yet explored. This comprehensive approach uncovers the multifaceted role of industrial big data use, assessing not only whether it independently influences product innovation but also if it acts as a mediator, enhancing the effects of sustainability orientation on innovation outcomes. The findings shift the perspective of industrial big data from their being a mere operational tool to their being a strategic asset that has the power to amplify the sustainability efforts of an organisation to produce innovation. The findings have significant implications for both theory and practice, providing guidance for firms in leveraging industrial big data use to enhance their innovation strategies within a sustainability framework. The paper is organised as follows. Section 2 reviews the relevant literature and argues the hypotheses posited. Section 3 explains the method used for testing the hypotheses, including data collection, measurement and analysis, and the results and findings are presented in Section 4. Section 5 discusses the results and section 6 concludes with a reflection on the implications of the findings, offering insights for both theory and practice in the field of sustainable manufacturing.

2. Theoretical background and hypotheses

This research is based on and aims to link two streams of literature. The first concerns the relationship between SO and product innovation, and the second relates to the relationship between big data use – or industrial big data, when available –and product innovation. Prior to linking them, the concept of SO is defined and discussed. Figure 1 at the end of the section depicts the conceptual model proposed.

2.1 The concept of sustainability orientation

Sustainability Orientation (SO), which is evolving as a paradigm in strategic management and entrepreneurship, has garnered increasing attention (Khizar et al. 2022). The concept is rooted in the integration of environmental and societal considerations into business operations (Kuckertz & Wagner, 2010). Definitions vary, but tend to share acknowledgment of the significance of environmental and social issues in managerial perceptions (Shou et al., 2019) and the readiness of organisations to implement sustainability initiatives (Tata & Prasad, 2015). Shepherd & Patzelt (2011) view SO as embracing objectives that preserve nature and support community life, aiming at both economic and non-economic gains. As Claudy et al. (2016) suggest, SO is part of a higher-order construct of strategic orientation, which also includes market, entrepreneurial and learning orientations.

Organisations typically adopt SO in response to stakeholder pressures, including demands from customers, employees and policymakers (Schaltegger et al., 2019; UN, 2015). Other drivers may be

the pursuit of long-term benefits such as sustainable competitiveness and economic gains derived from integrating sustainability principles (Calabrese et al., 2019; Khizar et al., 2021). The implementation of SO is not just a reactive measure but a strategic choice that aligns with the organisation's long-term objectives (Marco-Lajara et al., 2023).

Two dimensions are often considered in SO, namely sustainability culture and sustainability practices (Claudy et al., 2016). Sustainability culture refers to the organisational conventions, values, philosophies and beliefs that prioritise sustainability as a strategic norm (Adams et al., 2016; livari & livari, 2010). Practices involve including social and environmental considerations in operating strategies and procedures, and are further classified into internal practices, which are related to a company's daily operations, and external practices, pertaining to the broader supply chain (Ahmadi-Gh & Bello-Pintado, 2021; Laari et al., 2016; Wang & Dai, 2018). It could be argued that sustainability culture translates into sustainability practices. Internal practices, the focus of this study, relate to daily internal operations (Laari et al., 2016) and aim to lower ecological impacts through reduced pollution and resource use, and to address social concerns such as employee health, safety and their human rights (Dangelico & Pujari, 2010; Pujari, 2006).

2.2. The effect of sustainability orientation on product innovation

Product innovation, as defined by OECD & Eurostat (2018), involves new or significantly improved goods or services that diverge from previous offerings. Product innovations must significantly enhance one or more performance requirements or features, be it user utility, the addition of brandnew features or the enhancement of already existing ones. Quality, technical details, dependability, durability, economic effectiveness in usage, affordability, convenience, usability and user friendliness are all relevant functional attributes. Not all product improvements must enhance all performance requirements (OECD & Eurostat, 2018). Product innovation as a critical outcome of the new product development processes is essential for maintaining competitive advantage (Kleinschmidt et al., 2007; Marion & Fixson, 2021).

In the innovation literature, SO has often been related to new product development as a natural consequence of the argument that product innovation needs to contemplate sustainability at every stage, starting from the design phase. Some recent studies examine both the direct and the indirect connection between SO and product innovation from different perspectives (Ahmadi-Gh & Bello-Pintado, 2021; Cheng, 2020; Claudy et al., 2016; Du et al., 2016; Keskin et al., 2020; Obal et al., 2020; Zhao et al., 2021). However, this remains one of the least understood areas of sustainability management (Adams et al., 2016; Cheng, 2020; Claudy et al., 2016). Inconsistent empirical results in the literature are another indication of the need to explore the relationship between SO and product innovation. Some research has found positive impacts of SO on the success of new products and innovations (Du et al., 2016), while others have found detrimental effects (Sen & Bhattacharya, 2001). This section will argue the effect of SO on product innovation based on the natural resource-based view, as a development of the resource-based view.

According to the resource-based view, businesses with valuable, rare, inimitable and nonsubstitutable resources are viewed as having the capacity to generate a competitive advantage (Barney, 1991). As a progression from the resource-based view, the natural resource-based view (Hart, 1995) posits that companies can achieve a competitive advantage if they can overcome the obstacles related to the natural environment. These businesses can enhance their strategic skills to address environmental pollution, product stewardship and sustainable development issues, while simultaneously improving economic performance. This implies that they may gain a competitive advantage by implementing more environmentally friendly practices into their daily operations. As a result, SO implementation is argued to enable innovation managers to identify innovative solutions to ecological and social problems, resulting in operational efficiencies, higher quality products and greater value for customers (i.e., differentiation advantage), thereby increasing the likelihood of innovation success (Claudy et al., 2016; Porter, 2015). To some authors, a significant strategic instrument for enhancing a company's corporate reputation, achieving market differentiation and gaining first-mover advantages is the application of SO in the new product development process (Du et al., 2016; Zhao et al., 2021).

In the context of manufacturing, a firm's internal sustainability practices are considered important manufacturing practices to create value in multiple dimensions, such as improving product quality, resource efficiency, health, safety and other working conditions, and to lessen environmental burdens such as air and soil pollution, waste and resource consumption (Ahmadi-Gh & Bello-Pintado, 2021; Moldavska & Welo, 2017). In the same vein, using SO in the new product development process has been identified as a valuable strategic tool for enhancing a company's brand, achieving market differentiation and gaining first-mover advantages (Du et al., 2016). By applying SO practices, innovation teams may be able to eliminate inefficiencies in product disposal (e.g., harmful materials) and product use (e.g., longer useful product life), or in manufacturing processes (e.g., less process waste) (Fiksel, 2009). Eliminating such inefficiencies is expected to boost product innovation profit margins and return on innovation investments. Furthermore, increased sustainability can lead to higher quality (e.g., enhanced materials or safety) and/or reduced prices, improving customer value and driving new product sales (Claudy et al., 2016). Thus, we hypothesise the following:

H1: The implementation of SO *practices positively influences product innovation.*

2.3 The effect of industrial big data use on product innovation

Big data is a term describing the storage and analysis of large and/or complex datasets using a series of techniques, including but not limited to NoSQL, MapReduce and machine learning (Ward & Barker, 2013). Big data is often described as 3V: Volume of data, Velocity in its generation and Variety in its nature (Dubey et al., 2019; Kristoffersen et al., 2020; Zhou et al., 2014). Others describe it as 5V, adding Veracity and Value (Addo-Tenkorang & Helo, 2016; Zhong et al., 2016), or 6V, adding Variability (Trabucchi & Buganza, 2019; Wamba et al., 2015; Yudhistyra et al., 2020a). In their review, De Mauro et al. (2015, page 9) assert that "big data represents the information assets characterised by such a high volume, velocity and variety so as to require specific technology and analytical methods for its transformation into value."

Industrial big data, originating from diverse sources including devices, control units, robots and other equipment, is a subset of big data within industrial settings (Zhang et al., 2020) starting from the factory floor (Kirmse et al., 2019). This concept is derived from the broader term big data, which encompasses various data types and sources, including social media, environmental and consumer data (Kirmse et al., 2019). Industrial big data is a critical segment of the larger big data ecosystem, playing a vital role in enhancing industrial production scheduling, risk detection, condition monitoring, safety supervision and quality management, among other applications (Shaolin et al., 2021).

The effectiveness of big data is amplified by integrating the Internet of Things and Cloud technologies. This combination enables the interconnection of different equipment and systems, including production and customer management systems. Regular updates and analyses of data enhance decision-making processes, leading to improvements in manufacturing flexibility, product quality, energy efficiency and equipment maintenance (Rüßmann et al., 2015; Strange & Zucchella, 2017).

According to the resource-based view (Barney, 1991), a theoretical approach deemed suitable for examining big data and its impact on sustainability (Hazen et al., 2016) argues that while physical technology can be easily replicated, the strategic exploitation of technology using complex social resources can yield a competitive advantage (Barney, 1991). Nevertheless, the exploitation of technologies generally involves the use of socially complex resources to reach its potential, meaning that a firm can obtain a sustained competitive advantage if it can exploit the technology better than other firms. Big data applications with high potential in manufacturing include design, planning, material distribution and tracking, manufacturing process monitoring, quality control and equipment maintenance (Tao et al., 2018). Product design is particularly closely related to product innovation. In the era of big data, product design is evolving towards a data-driven approach, benefiting from industrial big data analytics (Kusiak & Salustri, 2007; Li et al., 2022). Innovative product features can emerge from integrating diverse data sources and using data-mining algorithms to reveal previously unseen value (Kusiak, 2009). For instance, to increase the difficult-to-attain quality of ceramic materials, a producer may monitor and enhance the machinery's functionality and the design of the finished product (Kusiak, 2017). Similarly, sensor-gathered data can identify product defects, leading to improvements in existing products (Niebel et al., 2019).

Recent studies show that predictive big data analytics positively influences product and process innovation (Saleem et al., 2020). In turn, prescriptive analytics, which involves complex mathematical models for business decision-making, has been linked to operational innovation and improved business performance (Aydiner et al., 2019). Big data has been shown to impact on new product development through new revenue creation and expansion of existing product lines (Addo-Tenkorang & Helo, 2016), and to enhance collaborative innovation (Feng et al., 2024) and supply chain innovation (Jaouadi, 2022). In summary, the effect of big data on product innovation has already been argued and found to be positive.

However, the challenge in manufacturing is not just the volume of data but also its variety and complexity. Industrial big data serve as the raw material for the information value chain, and the quality of these data is crucial for all subsequent processes. The relationship between industrial big data use and product innovation has not yet been tested, but as a particular type of big data, the same effect is hypothesised:

H2: The use of industrial big data positively influences product innovation.

2.4. The interplay between sustainable orientation, industrial big data use and product innovation

Having argued separately the positive expected effect of sustainable orientation and big data use on product innovation, this section explores the interplay among the three factors. Overall, it is assumed that big data use, as a valuable capability enabled by technology, can have an enhancing effect of SO practices on product innovation. This is so because firms focused on integrating economic, social and environmental challenges into their strategies are expected to place greater emphasis on exploiting the diverse opportunities provided by big data in the field of sustainability.

Big data technologies offer several advantages for sustainability. For example, while big data is widely regarded as one of the simplest methods for digitising the circular economy (Nobre & Tavares, 2017), it can also assist in the evaluation of cost-cutting initiatives such as remanufacturing (Ge & Jackson, 2014). Moreover, big data has been linked to improvements in energy and water management, waste recovery and recycling, and emissions reduction (Laskurain-Iturbe et al., 2021). Song et al. (2017) presented a study involving various papers that demonstrate different approaches to how big data can improve natural resource management, allowing natural resource consumption, energy efficiency, environmental efficiency and protection to be improved. Further, a study based on the Indian manufacturing industry supports these findings, indicating that big data and predictive analytics significantly contribute to social and environmental performance (Dubey et al., 2019).

According to the natural resource-based view, it is asserted that innovation teams are more likely to come up with novel solutions to ecological and social issues in organisations with a sustainability orientation, giving them a competitive advantage (Hart, 1995). There are additional hurdles when companies include sustainability in their innovation strategy because a firm's product innovation practices may be influenced by other agendas and steered in conflicting ways (Zhao et al., 2021). Given the intricate and multidimensional nature of integrating sustainability objectives into processes like new product development, scholars have suggested that the win-win paradigm proposed by the natural resource-based view may be oversimplified, while emphasising the significance of creating models that account for missing data, examining mediating mechanisms and assessing contextual factors (Ahmadi-Gh & Bello-Pintado, 2021; Akomea et al., 2022; Claudy et al., 2016; Roxas et al., 2017). In this direction, Zhao *et al.*, 2021 investigate the mediating role of 3D printing technologies in the link between SO and new product development processes, finding a mediation effect of 3D printing on product innovation. This result suggests that other technologies may also reduce the incidence of discrepancies between sustainability initiatives and product innovation. We argue that industrial big data use could have the potential to adequately handle these tensions.

In this line, when considering the influence of industrial big data on product innovation and its role in operationalising SO, it is plausible to hypothesise that industrial big data could serve as a mediating factor in the sustainability-product innovation nexus. In other words, industrial big data may be the key mechanism through which SO impacts product innovation. Based on these insights, this paper proposes the following hypothesis:

H3: The use of industrial big data positively mediates the effects of SO practices on product innovation.



Figure 1. The framework of hypotheses

3 Method

3.1 Data collection

The data for this study come from the European Manufacturing Survey (EMS), directed by the Fraunhofer Institute for Systems and Innovation Research (ISI) (Fraunhofer ISI, 2023). The survey collects detailed information on innovations in manufacturing. The main goal of the EMS initative is to learn more about the utilisation of production and information technology, the manufacture of organisational approaches and the adoption of best management practices. Integrating the most recent developments among the topics of interest enhances already existing innovation surveys.

The EMS survey was co-designed and conducted by the authors of this study, who are part of the EMS consortium. The data provide a first-hand, direct account of the subject matter, keeping the information closely linked to the event being studied, so the data can be defined as primary data for this study (Cohen et al., 2007). Other editions of EMS data have been used in papers related to innovation (Armbruster et al., 2008; Llach et al., 2012a; Llach et al., 2012b; Manresa et al., 2019; Pons et al., 2018; Serrano-García et al., 2022), sustainability and green manufacturing (Palčič et al., 2013; Pons et al., 2013; Šebo et al., 2021), and digitalisation (Blichfeldt & Faullant, 2021; Dachs et al., 2019; Lerch et al., 2022; Manresa et al., 2021; Vilkas et al., 2022).

The eight-page long survey with 25 sections is conducted among manufacturing companies (NACE Revision 2 codes from 10 to 33) with at least 20 employees. The EMS consortium uses various techniques to collect reliable data that allow for international comparison and the aggregation of different countries' datasets, following the recommendations endorsed by Survey Research Center (2011). These procedures are aimed at eliminating difficulties stemming from the use of multiple languages and national specialist terminology. To ensure uniformity, a basic questionnaire is first designed in English, translated into the national language and then subject to a reverse translation to check accuracy. This process involves translating the content intended for a specific language group back into the original source language. By comparing the original and re-translated versions in the source language, it becomes possible to identify and address any discrepancies or issues present in the translation for the target language audience. Pre-tests, which also help to detect possible translation mistakes, are undertaken in each participating country. Last, the same data harmonisation techniques are used (Bikfalvi et al., 2014).

The subsample of the EMS used in the present paper was collected in 2018. It consists of a total of 1,123 surveys, distributed as follows: Austria 253, Spain 85, Croatia 105, Lithuania 199, Slovakia 114, Slovenia 127 and Serbia 240. To meet the objectives of the present research in the innovation aspect, and given that the same questions and criteria for the sample selection were applied in each case, these countries' datasets were pooled, using as a criterion their ranking in the Global Innovation Index 2018. This index considers the most recent global innovation trends and ranks the innovation ecosystem performance of 132 economies, while highlighting innovation strengths and weaknesses and particular gaps in innovation metrics. The countries in the sub-sample selected for this study are in the second quartile in the ranking, indicating that they are among the most innovative and were chosen to facilitate a higher proportion of innovative firms in the database.

3.2 Measures

The study employs two primary constructs as independent variables: sustainability orientation (SO) and industrial big data (BD) use. The dependent variable is product innovation.

SO construct: the SO construct was calculated as the sum of different sustainability practices used in manufacturing firms, as listed in Table 1. Internal sustainability practices include a variety of environmental and socially responsible actions that a company adopts and puts into practice throughout its internal business operations. Environmental practices aim to reduce pollution and the use of both natural resources and conventional sources of energy, and include water efficiency

management (Ahmadi-Gh & Bello-Pintado, 2021; Montabon et al., 2007), energy management (Montabon et al., 2007), environmental certification (Montabon et al., 2007; Wang & Dai, 2018) and re-use and recycling (Ahmadi-Gh & Bello-Pintado, 2021; Montabon et al., 2007; Wang & Dai, 2018). Internal practices include social, responsible management practices focused on emplyees' health, safety and human rights, including employment training and programmes to promote staff development and loyalty (Montabon et al., 2007; Wang & Dai, 2018). Since seven different practices were considered, the construct has values from 0 to 7, with each practice given a value of 1 when marked as used by respondent companies.

BD construct: the BD construct was composed of the different purposes for which companies employed collected data, to evaluate the intensity of their use. Since six different uses were evaluated, the construct had values from 0 to 6, with each use given a value of 1. The value of the construct was zero if data were not collected, or if collected data were not used at all. As for the different data uses, the respondents could choose from optimising production processes, planning maintenance and repairs, planning resource utilisation and application, preparing productivity or key performance indicators, risk analysis and other uses.

Dependent Variable: Product Innovation was obtained from the answer to the following survey question: 'Has your factory introduced products since 2015 that were new to your factory, or incorporated major technical improvements? (e.g., use of new materials, modifications of product function, changes in operating principles etc.)'. Given that the possible answers were either yes or no, this is a dichotomic variable.

Descriptive statistics for the dependent and independent variables are detailed in Table 1, indicating that most of the firms in the sample had launched new products since 2015 (64%) and, with an average of 2.23 out of a maximum of seven, that the firms applied a relatively small number of SO practices. Furthermore, the number of different uses of industrial BD was rather low, at an average of 1.66 out of a maximum of six. Frequencies for the SO and BD constructs are shown in

Figure 2.

Table 1. Descriptive statistics for the dependent and independent variables

		N	Min	Max	Mean	Std. Deviation
Dependent variable						
Product innovation		1,102	0	1	0.64	0.48
Independent variable	25					
SO construct		1,123	0	7	2.24	1.46
	Organisational concepts – certified energy management system	1,048	0	1	0.12	0.33
	Organisational concepts – certified environmental management system	633	0	1	0.41	0.49
	Human resources – instruments to promote staff loyalty	1,077	0	1	0.57	0.5
	Human resources – training on the job	1,074	0	1	0.71	0.45
	Energy efficiency technologies – water re-use and recycling	1,036	0	1	0.24	0.43
	Energy efficiency technologies – kinetic and process energy recuperation	1,044	0	1	0.25	0.43
	Product related services – take-back services (recycling, disposal, taking back)	1,055	0	1	0.22	0.41
BD construct		1,123	0	6	1.66	1.79
	Use of machines/systems that store operating data	1105	0	1	0.62	0.49
	Optimisation of production processes	685	0	1	0.62	0.49
	Planning maintenance and repairs	685	0	1	0.41	0.49
	Planning resource utilisation and application	685	0	1	0.41	0.49
	Preparing KPIs	685	0	1	0.55	0.5
	Risk analysis	685	0	1	0.58	0.49
	Other use	685	0	1	0.15	0.36
	No use of data	685	0	1	0.21	0.41



Figure 2 Frequencies of SO and BD constructs

Control Variables: In line with extant research, general resource availability, openness and sector innovation intensity were considered as control variables. The resources available to a company have a significant impact on its capacity to achieve its goals via various activities, including innovation-related ones. Resource availability will be considered in two dimensions, size and sector. The most common measure for resource availability is firm size, which is commonly considered a driver of innovation activities and an indication of a company's willingness to innovate (Cohen & Klepper, 1996). The most common measures of firm size include the number of people employed and the volume of turnover. The Oslo Manual 2018 suggests using both. Due to the fact that the number of employees does not follow a normal distribution, a transformation is required using parametric statistics (Tabachnick et al., 2007) and the natural logarithm of the original values, resulting in a normal distribution. Employees and turnover values are for 2018.

As regards openness, when businesses operate on a global scale, they develop in-depth knowledge of and acute responses to institutional settings, influencing new product performance (Zhao et al., 2021). This explains why export share and import share are included as control variables. In turn, innovation intensity is a control variable that classifies industries into five groupslow, medium-low, medium, medium-high and high innovative sectors, according to the Peneder taxonomy of innovative intensity (Peneder, 2010). Sector innovation intensity is used as a proxy for the differential innovativeness of different sectors, as is common in previous innovation research (Manresa et al., 2018; Marques et al., 2015). Descriptive statistics for the selected control variables are presented in Table 2.

Size	Frequencies	%			
less than 50	510	46%			
between 51 and 249	447	40%			
more than 250	150	14%			
Country					
Austria	253	23%			
Spain	85	8%			
Croatia	105	9%			
Lithuania	199	18%			
Slovakia	112	10%			
Slovenia	127	11%			
Serbia	240	21%			
Sector Innovation intensity					
Low	78	7%			
Medium-low	170	15%			
Medium	213	19%			
Medium-high	249	23%			
High	400	36%			
Turnover millions					
EUR					
less than 2	245	25%			
between 2 and 10	370	38%			
between 1 and 50	255	26%			
more than 50	117	11%			

Table 2. Sample distribution of companies by size, and sector innovation intensity

3.3 Data analysis

Because of the binomial nature of the dependent variable, a binomial logistic regression was used to test the hypotheses relating to the total effect of the independent variables on the dependent variable (H1 and H2). H3 was tested using the mediation model proposed by Hayes (2018). SPSS Version: 28.0.0.0 (190). The control function approach is used to address endogeneity, with the residuals of the firststage regression added as an additional regressor into the main regression (Ebbes et al., 2016; Petrin & Train, 2010). The idea is that the residuals capture the omitted variables that make independent variables endogenous. By including this term in the main regression, we control for endogeneity (Ebbes et al., 2016). For the first regression, market-related aspects were considered as the dependent variable, measured via the R+D market construct, composed of R+D shared with clients/suppliers and R+D shared with other companies. As the independent variable for the regression model, the construct SO was used for the first model and BD for the second model. The standardised residuals from this regression model were then used in the main logistic regression model.

Several procedural and statistical remedies were adopted to diminish common method bias (Podsakoff et al., 2003; Rodríguez-Ardura & Meseguer-Artola, 2020). First, the respondents were guaranteed that their answers were anonymous. Second, more than one single respondent was allowed to answer the survey, adding more information sources to gather data about constructs in the model. Third, Harman's single-factor test was performed, the result indicating that one factor explains only 17.03% of the variance, and that this factor did not capture most of the variance. These results therefore suggest that common method bias is not an issue in this study (Tehseen et al., 2017).

4 Results

4.1 Descriptive statistics

Figure 3 shows the adoption of sustainability orientation practices (SO practices) in European manufacturing firms. The predominant category of SO practices is the one related to human resources. Training on-job is clearly the most adopted sustainability practice, featuring in more than 70% of the companies, followed by instruments to promote staff loyalty. This could be due to the

fact that the aforementioned practices were adopted earlier than the other analysed practices. Even though 25 % of the companies implemented technologies to recover energy, only 12% of them are certified for the energy management system. Environmental management certification is much more implemented, present in 41% of the companies. On analysing the median for the year when these systems were implemented, a clear advantage for the chronologically older available certifications can be seen: 2011 for the environmental system and 2014 for the energy system. Nevertheless, looking at their plans for implementation, 19% of the companies want to implement environmental systems and only 15% energy systems, suggesting that certified energy system levels will not reach environmental system levels in the coming years. Contrarily, the difference between their implementation will be increasingly pronounced.



Figure 3. Implementation of SO practices

Technologies for recycling and water re-use, together with technologies to recover kinetic and process energy, have values of 24% and 25%, respectively, showing that they are not frequently employed in the companies. The forecast for the next three years shows that these proportions will not grow significantly: only 8% of the companies plan to adopt technologies for water treatment, and 11% for energy treatment. Both technologies have been updated in the adopting companies in the last three years by an average of 30%.

Figure 4 shows the extent to which the companies estimate the implementation of sustainability practices. The question in the survey asks respondents to evaluate the potential of use as low, medium or high on a scale of 1 (low potential) to 3 (high potential), meaning the extent to which the companies believe that sustainable practices are deployed inside the company. The data reveal that potential of use is highest in applying principles of environmental and energy certification. Contrarily, it appears that instruments to promote staff loyalty and technologies for energy recuperation could be better used.



Figure 4. Potential of use of SO practices

The findings show that the rates of implementation of the studied internal environmental SO practices are relatively low in European manufacturing firms, ranging from 12% for certified energy management systems to 41% for certified environmental management systems. The level of social SO practices is higher, with percentages ranging from 57% to 71%. As for the intensity of use of the SO practices, companies state medium intensity use, with an average value of 2.26 on a scale of 1 to 3, clearly indicating that the practices are not being used to their full potential and there is room for growth. This untapped potential growth may be the sum of a number of factors, including the state

of the economy, lack of expertise, managers believing that certain practices are unneeded given the modest size of the business, and so on.

Industrial big data use appears to be widely implemented among the companies (Figure 5): 62% use machines or systems that automatically store operating data. Companies use these data for different purposes, but mostly for optimising the production processes. However, 21% of the companies do not use the collected data at all, even though they are collecting them. There are different possible explanations for this, including insufficient technology and techniques to capture value from industrial big data, a lack of talent or incentives to make use of the data, or a lack of competitive intensity or present need to leverage the benefits of industrial big data.



Figure 5. Use of industrial big data

When evaluating product innovation, new products were introduced in an average of 64% of the companies. Interestingly, product innovation is not growing in line with the level of sector innovation intensity. The group of companies where the proportion of new products is the highest coincides with the high innovation intensity sector, while the group of companies with less product innovators are categorised as medium intensity innovators as opposed to low intensity innovators, as seen in Figure 6.



Figure 6. Product innovation by sector innovation intensity.

4.2 Hypotheses testing

To test the separate relationship between SO practices and industrial big data use in product innovation (H1 and H2, respectively), two sets of binary regression logistic models were performed. The results are given in Table 3.

The dependent variable in all cases was the product innovation measured as new-to-the-firm products (new products in the last three years), with values of 0 for no product innovation and 1 for new products for the company. First, a model with control variables was tested. The method used to obtain the control variables that are significant for the model was a binary logistic regression with backward Ward analysis. The variables entered initially were the number of employees, imports, exports, sector of innovation intensity and annual turnover. The backward Ward analysis was performed in four steps until the control variables with statistical significance for the model were determined. In the end, only the variables number of employees and imports proved to be relevant for the model. They were therefore the variables used for the following models to test the hypotheses. The first model tested hypothesis H1. The construct of SO together with significant control variables were entered as independent variables. The model showed that the SO has significance in product innovation with a value p<0.10 (sig. 0.064) and a β of 0.094, providing support for the first hypothesis.

The second model tested hypothesis H2, using the industrial big data construct as the independent variable, together with the significant control variables. This model showed that the construct industrial big data has statistical significance with a value p<0.050 (sig. 0.015) and a β of 0.118, providing support for H2, that industrial big data positively influences product innovation.

Table 3. Regression models for total effect of SO and BD on product innovation

	Model I.		Model II.	
Dependent variable	Product	innovation		
Independent variable	β	sig.	β	sig.
SO construct	0.094	0.064		
BD construct			0.118	0.015
Control variables				
Imports	0.005	0.01	0.005	0.014
Market Residuals	0.431	<0.001	0.425	<0.001
Number of employees	0.237	0.002	0.239	0.002
Number of analysed cases	913		913	
R ² Cox & Snell	0.076		0.077	
R ² Nagelkerke	0.103		0.105	

To test H3, the model used was the mediation model, which focuses on the estimation of the indirect effect of the SO construct on product innovation through an intermediary mediator variable, in this case the BD construct (Hayes, 2017).

The path of the direct effect from SO to BD (a) is positive and statistically significant (β = 0.48, p < 0.001). The path of the direct effect from BD to product innovation (b) is also positive and statistically significant (β = 0.082, p < 0.05). The path of the direct effect from SO to product innovation is positive, but insignificant (β = 0.056, p > 0.05). The indirect effect was tested using bootstrapping (5,000 re-samples in the model, 1-tailed significance), a nonparametric resampling

procedure that does not impose the assumption of normality on the sampling distribution (Preacher & Hayes, 2008). If the zero falls outside the lower limit (LL) and the upper limit (UL) of the 95% confidence interval (Cl), then the indirect effect is significant. Examining the indirect or mediating effect of BD on the relationship between SO and product innovation, calculated by a^*b , the result is $\beta = 0.0391$, and this result is statistically significant at 95% CI = (Boot LLCI =0.001, Boot ULCI= 0.079). Therefore, the model suggests that BD has a positive mediating effect on the relationship between SO and product innovation.

The summary of the Hayes mediation model, calculated by PROCESS.SPS software for SPSS, is shown in table 4.

Tahle 4. Ha	ves PROCESS SPS mediation	model
10010 4.110	yes i nocessisi s meanation	mouci

Sample size: 1040										
Consequent: BD					Consequent: product innovation					
	Model A	: SO-> BD	BD			Model B	Model B: BD-> Innovation			
	R	R-sq				Cox & Snell	Nagelkerne			
	0.409	0.167				0.051	0.069			
Variables	β	SE	р	LLCI	ULCI	β	SE	р	LLCI	ULCI
SO	0.478	0.353	0.000	0.409	0.547	0.056	0.050	0.263	-0.042	0.154
BD						0.082	0.041	0.047	0.001	0.162
import	0.005	0.002	0.004	0.002	0.008	0.006	0.002	0.002	0.002	0.010
number of	0.099	0.508	0.052	-0.001	0.199	0.323	0.714	0.000	0.183	0.463
employees										
Direct and inc	direct effect	cts								
Direct effect o	of SO on pr	oduct innov	vation			-				
	effect	SE	р	LLCI	ULCI					
	0.056	0.050	0.263	-0.042	0.154					
Indirect effect of SO on product innovation through BD										
	effect	Boot SE	LLCI	ULCI		_				
BD	0.039	0.020	0.001	0.079						

The final research models with the above-described results of the analysis are presented in Figure



Figure 7. Research models with results

The results of the tested hypothesis can be found in Table 5.

Table 5. Summary of hypotheses testing

Hypothesis	Effect	Model	Result
H1	+	SO -> Product innovation	Supported
H2	+	BD -> Product innovation	Supported
H3	+	SO -> BD - > Product innovation	Supported

In summary, these results suggest that SO practices and industrial big data use positively influence product innovation. Industrial big data use positively mediates in the relationship between SO and product innovation, thereby helping to achieve a positive influence of SO on new products. The hypotheses are designed to capture the multifaceted influence of industrial big data within the context of SO and product innovation.

5. Discussion

5.1 Sustainability orientation direct impact on product innovation

The findings relating to Hypothesis 1 reveal a compelling narrative in favour of SO because of its positive influence on product innovation, adding further support to some existing findings (Claudy et al., 2016). The support for this hypothesis confirms that sustainability is not merely a compliance or ethical imperative but that it also serves as fertile ground for innovation.

Our findings particularly support that SO spurs innovation. The direct relationship between SO and product innovation can be explained by means of several mechanisms. First, an orientation towards sustainability often necessitates creative thinking and the reimagining of product life cycles, which inherently fosters innovation. Second, this orientation typically requires the adoption of new processes and technologies. Last, by adopting SO, firms can attract talent and collaborate with stakeholders who share similar values, thereby enhancing their innovative capacity.

5.2 The direct impact of industrial big data use on product innovation

While previous research has studied the positive effect of digital technologies on product innovation (Zhao et al., 2021), our findings enlarge this knowledge for the particular case of industrial big data. The substantiation of H2 provides evidence of the direct effect of industrial big data use, gathered from machines and systems in the production process, on product innovation within the manufacturing sector. This relationship is illustrative of the potential of the digital age to revolutionise traditional business models and processes. Firms that effectively utilise industrial big data are shown to be significantly more innovative, which suggests that the capacity to use and analyse large datasets is a critical driver of product development and competitive advantage.

The impact of industrial big data use on product innovation may manifest in several ways. It allows for the optimisation of the product development process by predicting potential failures and successes, thereby reducing costs and time-to-market. It also empowers companies to engage in

predictive analytics, enabling them to foresee future trends and prepare innovative responses in advance.

This direct impact of industrial big data use on innovation also illustrates the shifting paradigms in product development. No longer is innovation confined to the R&D laboratories, but it is now increasingly taking place at the intersection of technology, data science and strategic business analysis. Firms that invest in industrial big data capabilities are positioned at the forefront of this shift, ready to capitalise on the opportunities presented by the data-driven economy. Our results are therefore consistent with Saleem et al. (2021), although we explore a further synergetic effect of industrial big data, as presented in the next section. Direct effects of the mediating variable on the dependent variable, and not only its mediating effects, have also been examined in previous studies related to sustainability and digitalisation (Črešnar et al., 2023; Rehman et al., 2022).

5.3 The mediating role of industrial big data use on product innovation

H3 delves into the intricacies of the mediating effect of industrial big data use on the relationship between SO and product innovation. This hypothesis is of particular interest as it reflects the transformative power of industrial big data in terms of actualising the principles of sustainability in a tangible and impactful way. The confirmation of this hypothesis underscores the pivotal role that industrial big data use plays in translating sustainability intentions into innovative outcomes.

The analysis supports the idea that industrial big data use acts as a critical intermediary that takes the intentions of SO and operationalises them into data-driven strategies for product innovation. While SO sets the strategic direction for innovation, industrial big data use provides the means to achieve it. The use of industrial big data enables firms to integrate sustainability into the product innovation process more effectively, connecting the gap between sustainability goals and marketable products. For instance, big data use can provide insights into sustainable materials and processes or help optimise supply chains for reduced environmental impact. The mediating role of industrial big data also suggests that while SO can independently promote product innovation, the depth and breadth of this innovation are significantly enhanced when underpinned by data use. Firms can quantify and analyse their sustainability efforts, translate sustainability goals into specific innovation targets, and monitor the success of these initiatives through industrial big data use. Industrial big data thereby acts as the bridge that connects the conceptual (SO) with the practical (product innovation), enabling firms to implement their sustainability orientation in a concrete and measurable way. This mediation effect is particularly relevant in the context of the European manufacturing sector, where regulatory pressures and market dynamics demand both innovation and sustainability.

In essence, the validation of H3 enriches our understanding of the symbiotic relationship between sustainability orientation and industrial big data use to foster product innovation, which to the author's knowledge is new in the literature. It lays the groundwork for a more integrative approach to innovation strategy that leverages the power of industrial big data to fulfil the promise of sustainability in the competitive landscape of manufacturing.

6. Conclusion

This research explores the complex interplay between sustainability orientation (SO), industrial big data use and product innovation within the European manufacturing sector. We found robust support for all the proposed hypotheses, highlighting the dual role of industrial big data use as both an independent and a mediating variable. The simultaneous support for these hypotheses does not signify inconsistency but rather reinforces the complexity of the relationships at play. It highlights the comprehensive impact of big data on the innovation ecosystem, both as a standalone influence on product innovation and as an intermediary that enhances the effect of SO on innovation. This dual capacity of big data use aligns with the broader research objective of understanding its multifaceted role. When considering the hypotheses in a joint context, it becomes clear that SO and industrial big data use are not mutually exclusive in their impact on product innovation. Instead, they operate in a synergistic manner. SO provides the vision and impetus for innovation, industrial big data use equips firms with the tools to realise this intention, and the innovation outcomes are the manifestations of this combined effect. Furthermore, industrial big data use does not merely support SO-driven innovation, but it also independently stimulates innovation by revealing new opportunities and offering the analytics capability to capitalise on them.

6.1 Theoretical contributions

First, this study adds knowledge to the ongoing debate about the relationship between SO and product innovation. Drawing on the natural resource-based perspective, a company's SO is a strategic orientation that is represented in its activities and culture (Hart, 1995). It was therefore postulated that implementing an SO at the organisational level would help new product development managers find creative solutions to ecological and social issues, which would lead to increased operational effectiveness, better product quality and greater customer value, ultimately increasing the likelihood of product innovation success (Porter & Claas van der Linde, 1995). Based on the results obtained, the models in this study indicate that the implementation of SO practices themselves presents a significant and positive relationship with product innovation. This contributes to the theoretical arguments that emphasise the positive effect of SO on product innovation, including the study by Claudy et al., (2016), which implies that implementing an SO has a beneficial impact on new product development performance. Moreover, our results suggest that the adoption of SO practices has a positive effect on product innovation regardless of innovation intensity sector, which has been scarcely considered in previous research.

Second, the study confirms the positive effect of industrial big data use on product innovation, which aligns with the resource-based view in that it identifies big data as a strategic resource that can directly fuel innovation. The capability to collect, analyse and interpret large volumes of data is a critical asset that enables firms to stay ahead in the innovation race. This capability is especially pertinent in today's data-intensive business environments where making informed, data-driven decisions is key to developing products that meet evolving market needs and staying ahead of competitors. This finding contributes to the literature on the role of digital technologies in product innovation, a relationship that has been relatively underexplored. The results of the study align with Saleem et al. (2021), who also observed a positive effect of big data from various sources on product innovation.

Third, an important contribution of this research is the identification of the route through which SO is related to product innovation. The results suggest that industrial big data use mediates in the innovation outcome of SO, extending the sustainability literature by uncovering a previously unrecognised relation of big data use and SO. This finding supports the logic that big data use has the potential to adequately handle tensions between the economic, social and environmental goals of sustainability—the well-known triple bottom line (Elkington & Rowlands, 1999), where firms may run into difficulties when implementing SO in the creation of new goods.

6.2 Practical contributions

In recent years, the focus on sustainability has become a significant concern for management boards, driven by societal and environmental demands. Many businesses are integrating sustainability into their models, viewing it as essential for maintaining or improving their reputation and brand value. This includes developing sustainability policies, managing carbon footprints and selecting suppliers with sustainability standards. However, there is still uncertainty among managers about the tangible benefits of these sustainable practices.

This study reveals that sustainability practices positively impact product innovation, although their implementation and potential exploitation remain limited in European manufacturing companies. Managers are encouraged to not only adopt these practices but also to intensify their application. Senior executives are advised to approach sustainability as a strategic priority, requiring advanced techniques, technologies and effective data utilisation to enhance product innovation.

The research indicates a stronger link between sustainability and product innovation when data are applied across various functions, such as performance indicators, production optimisation, risk analysis, resource planning and maintenance. The study highlights the importance of using big data in diverse applications, beyond just data collection and storage.

The results emphasise the need for a sustainability orientation that is actionable through industrial big data use. Policymakers should focus on promoting big data technology adoption as a driver of sustainable innovation. Investing in big data technologies, aligning the sustainability goals with IT strategies and cultivating data-analytic skills within the workforce are crucial for maintaining and enhancing innovation capacities. For theorists, these findings beckon a re-evaluation of innovation models to incorporate the role of data analytics as a fundamental component.

6.3 Limitations and future research

Using only manufacturing companies for the research is advantageous for acquiring a deeper knowledge about this particular sector. However, it is also a limitation, and scholars could examine other sectors to see if the relationships found are likewise valid.

The number of sustainability practices available for the study could be considered a limitation, so future analyses should include other SO practices to enrich the results. While the results uncovered the importance of internal sustainability practices, the research would be enriched by including external sustainability practices and sustainability culture, although it can be assumed that culture drives practices, as studied in this research. As regards the dependent variable, the authors were only able to assess new-to-firm product innovation, while ecological and social factors of innovated products were not considered (Fiksel, 2009). Future studies could evaluate how SO affects the social, environmental and economic facets of product innovation.

This study examines how industrial big data use mediates the relationship between SO and product innovation. In this regard, it may be advantageous to look at other important new technologies to resolve potential conflicts in sustainability programmes that support product development and innovation. It is conceivable that factors that were not considered in this study could influence both SO and product innovation, although these factors have not been consistently identified in the extant literature. The capacity to collect all potentially pertinent data was constrained by the data set used in this study, despite providing exclusive access to a wide range of data. Therefore, this study could be seen as a supplement to the body of literature, and as such future research could build on this basic foundation.

Our study revolves aroung the use of industrial big data, recognising that big data encompasses a broader concept. Future research could explore other types of big data to expand upon our findings, such as consumer and environmental data (Kirmse et al., 2019). Further, future studies are encouraged to delve into the specific uses of big data that catalyse innovation within different organisational contexts and sectors. This could encompass areas such as big data analytics in service innovation, the utilisation of big data in improving the innovation lifecycle, and the role of big data in fostering open innovation and collaborative ventures. Each application offers a unique perspective on how big data can be leveraged to stimulate innovation.

This article specifically focuses on product innovation within the context of sustainability orientation and its impact on the innovation process. However, it is important to acknowledge that innovation encompasses a broader spectrum than just product innovation. Other forms of innovation such as process, marketing and organisational innovations also play a crucial role in shaping businesses and industries. While our discussion is centered on product innovation, we recognise the importance of exploring how sustainability orientation influences the other types of innovation. This recognition not only broadens the scope of our inquiry but also encourages further exploration and studies in these areas, opening new avenues for future research, particularly within the framework of the natural resource-based view theory, thereby enriching our understanding of innovation in its various forms and its relationship with sustainability.

Despite the limitations, the results of the study enrich our understanding of the symbiotic relationship between sustainability orientation and industrial big data use to enhance product innovation, laying the groundwork for a more integrative approach to innovation strategy. This research opens avenues for further investigation into how different aspects of sustainability orientation (such as environmental management, social responsibility, etc.) specifically contribute to product innovation.

Data statement

Due to the sensitive nature of the questions asked in this study, survey respondents were assured that raw data would remain confidential and would not be shared.

Data not available / The data that has been used is confidential.

References

- Abdul-Rashid, S. H., Sakundarini, N., Raja Ghazilla, R. A., & Thurasamy, R. (2017). The impact of sustainable manufacturing practices on sustainability performance: Empirical evidence from Malaysia. *International Journal of Operations & Production Management*, *37*(2), 182–204. https://doi.org/10.1108/IJOPM-04-2015-0223
- Adams, R., Jeanrenaud, S., Bessant, J., Denyer, D., & Overy, P. (2016). Sustainability-oriented Innovation: A Systematic Review: Sustainability-oriented Innovation. *International Journal of Management Reviews*, *18*(2), 180–205. https://doi.org/10.1111/ijmr.12068
- Addo-Tenkorang, R., & Helo, P. T. (2016). Big data applications in operations/supply-chain management: A literature review. *Computers & Industrial Engineering*, *101*, 528–543. https://doi.org/10.1016/j.cie.2016.09.023

- Ahmadi-Gh, Z., & Bello-Pintado, A. (2021). The Effect of Sustainability on New Product Development in Manufacturing—Internal and External Practices. *Administrative Sciences*, *11*(4), 115. https://doi.org/10.3390/admsci11040115
- Akomea, S. Y., Agyapong, A., Ampah, G., & Osei, H. V. (2022). Entrepreneurial orientation,
 sustainability practices and performance of small and medium enterprises: Evidence from an
 emerging economy. *International Journal of Productivity and Performance Management*.
 https://doi.org/10.1108/IJPPM-06-2021-0325
- Armbruster, H., Bikfalvi, A., Kinkel, S., & Lay, G. (2008). Organizational innovation: The challenge of measuring non-technical innovation in large-scale surveys. *Technovation*, 28(10), 644–657. https://doi.org/10.1016/j.technovation.2008.03.003
- Aupperle, K., Hatfield, J. D., & Carroll, A. B. (1983). Instrument Development and Application in
 Corporate Social Responsibility. *Academy of Management Proceedings*, *1983*(1), 369–373.
 https://doi.org/10.5465/ambpp.1983.4976378
- Aydiner, A. S., Tatoglu, E., Bayraktar, E., Zaim, S., & Delen, D. (2019). Business analytics and firm performance: The mediating role of business process performance. *Journal of Business Research*, *96*, 228–237. https://doi.org/10.1016/j.jbusres.2018.11.028

Barney, J. (1991). Firm Resources and Sustained Competitive Advantage. 23.

- Bashtannyk, V., Buryk, Z., Kokhan, M., Vlasenko, T., & Skryl, V. (2020). Financial, economic and sustainable development of states within the conditions of industry 4.0. *International Journal of Management*, *11*(4), 406–413. Scopus.
 https://doi.org/10.34218/IJM.11.4.2020.040
- Bikfalvi, A., Jäger, A., & Lay, G. (2014). The incidence and diffusion of teamwork in manufacturing evidences from a Pan-European survey. *Journal of Organizational Change Management*, 27(2), 206–231. https://doi.org/10.1108/JOCM-04-2013-0052

Blichfeldt, H., & Faullant, R. (2021). Performance effects of digital technology adoption and product
 & service innovation – A process-industry perspective. *Technovation*, *105*, 102275.
 https://doi.org/10.1016/j.technovation.2021.102275

- Calabrese, A., Costa, R., Levialdi, N., & Menichini, T. (2019). Integrating sustainability into strategic decision-making: A fuzzy AHP method for the selection of relevant sustainability issues. *Technological Forecasting and Social Change*, *139*, 155–168. https://doi.org/10.1016/j.techfore.2018.11.005
- Chatterjee, S., Chaudhuri, R., Shah, M., & Maheshwari, P. (2022). Big data driven innovation for sustaining SME supply chain operation in post COVID-19 scenario: Moderating role of SME technology leadership. *Computers & Industrial Engineering*, *168*, 108058. https://doi.org/10.1016/j.cie.2022.108058
- Cheng, C. C. J. (2020). Sustainability Orientation, Green Supplier Involvement, and Green Innovation Performance: Evidence from Diversifying Green Entrants. *Journal of Business Ethics*, *161*(2), 393–414. https://doi.org/10.1007/s10551-018-3946-7
- Ciampi, F., Demi, S., Magrini, A., Marzi, G., & Papa, A. (2021). Exploring the impact of big data analytics capabilities on business model innovation: The mediating role of entrepreneurial orientation. *Journal of Business Research*, *123*, 1–13.

https://doi.org/10.1016/j.jbusres.2020.09.023

Claudy, M. C., Peterson, M., & Pagell, M. (2016). The Roles of Sustainability Orientation and Market Knowledge Competence in New Product Development Success: M. C. CLAUDY ET AL. *Journal* of Product Innovation Management, 33, 72–85. https://doi.org/10.1111/jpim.12343

Cohen, L., Manion, L., & Morrison, K. (2007). *Research methods in education* (6th ed). Routledge.

Cohen, W. M., & Klepper, S. (1996). Firm Size and the Nature of Innovation within Industries: The Case of Process and Product R&D. *The Review of Economics and Statistics*, *78*(2), 232. https://doi.org/10.2307/2109925

- Črešnar, R., Dabić, M., Stojčić, N., & Nedelko, Z. (2023). It takes two to tango: Technological and nontechnological factors of Industry 4.0 implementation in manufacturing firms. *Review of Managerial Science*, *17*(3), 827–853. https://doi.org/10.1007/s11846-022-00543-7
- Dachs, B., Kinkel, S., & Jäger, A. (2019). Bringing it all back home? Backshoring of manufacturing activities and the adoption of Industry 4.0 technologies. *Journal of World Business*, *54*(6), 101017. https://doi.org/10.1016/j.jwb.2019.101017
- Dangelico, R. M., & Pujari, D. (2010). Mainstreaming Green Product Innovation: Why and How
 Companies Integrate Environmental Sustainability. *Journal of Business Ethics*, 95(3), 471–
 486. https://doi.org/10.1007/s10551-010-0434-0
- De Mauro, A., Greco, M., & Grimaldi, M. (2015). What is big data? A consensual definition and a review of key research topics. 97–104. https://doi.org/10.1063/1.4907823
- Du, S., Yalcinkaya, G., & Bstieler, L. (2016). Sustainability, Social Media Driven Open Innovation, and New Product Development Performance*: SUSTAINABILITY, OPEN INNOVATION, AND NPD. *Journal of Product Innovation Management*, *33*, 55–71. https://doi.org/10.1111/jpim.12334
- Dubey, R., Gunasekaran, A., Childe, S. J., Papadopoulos, T., Luo, Z., Wamba, S. F., & Roubaud, D.
 (2019). Can big data and predictive analytics improve social and environmental sustainability? *Technological Forecasting and Social Change*, *144*, 534–545. https://doi.org/10.1016/j.techfore.2017.06.020
- Ebbes, P., Papies, D., & van Heerde, H. J. (2016). Dealing with Endogeneity: A Nontechnical Guide for Marketing Researchers. In C. Homburg, M. Klarmann, & A. Vomberg (Eds.), *Handbook of Market Research* (pp. 1–37). Springer International Publishing. https://doi.org/10.1007/978-3-319-05542-8_8-1
- Elkington, J., & Rowlands, I. H. (1999). Cannibals with forks: The triple bottom line of 21st century business. *Alternatives Journal*, *25*(4), 42.

Fiksel, J. (2009). *Design for Environment: A Guide to Sustainable Product Development* (2nd ed.). McGraw-Hill Education.

https://www.accessengineeringlibrary.com/content/book/9780071605564

Fisher, O. J., Watson, N. J., Escrig, J. E., Witt, R., Porcu, L., Bacon, D., Rigley, M., & Gomes, R. L.
(2020). Considerations, challenges and opportunities when developing data-driven models for process manufacturing systems. *Computers & Chemical Engineering*, 140, 106881. https://doi.org/10.1016/j.compchemeng.2020.106881

Fraunhofer ISI. (2023). European Mnaufacturing Survey.

https://www.isi.fraunhofer.de/en/themen/wertschoepfung/fems.html

- Ge, X., & Jackson, J. (2014). The Big Data Application Strategy for Cost Reduction in Automotive Industry. SAE International Journal of Commercial Vehicles, 7(2), 588–598.
 https://doi.org/10.4271/2014-01-2410
- Hallstedt, S. I., Thompson, A. W., & Lindahl, P. (2013). Key elements for implementing a strategic sustainability perspective in the product innovation process. *Journal of Cleaner Production*, *51*, 277–288. https://doi.org/10.1016/j.jclepro.2013.01.043
- Hallstedt, S., Isaksson, O., & Öhrwall Rönnbäck, A. (2020). The Need for New Product Development Capabilities from Digitalization, Sustainability, and Servitization Trends. *Sustainability*, *12*(23), 10222. https://doi.org/10.3390/su122310222
- Hämäläinen, E., & Inkinen, T. (2019). Industrial applications of big data in disruptive innovations supporting environmental reporting. *Journal of Industrial Information Integration*, *16*, 100105. https://doi.org/10.1016/j.jii.2019.100105
- Hart, S. L. (1995). A Natural-Resource-Based View of the Firm. *Academy of Management Review*, 20(4), 986–1014. https://doi.org/10.5465/amr.1995.9512280033
- Hart, S. L., & Ahuja, G. (1996). DOES IT PAY TO BE GREEN? AN EMPIRICAL EXAMINATION OF THE RELATIONSHIP BETWEEN EMISSION REDUCTION AND FIRM PERFORMANCE. *Business*

Strategy and the Environment, 5(1), 30–37. https://doi.org/10.1002/(SICI)1099-0836(199603)5:1<30::AID-BSE38>3.0.CO;2-Q

- Hayes, A. F. (2017). Introduction to mediation, moderation, and conditional process analysis: A regression-based approach. Guilford publications.
- Hayes, A. F. (2018). Partial, conditional, and moderated moderated mediation: Quantification, inference, and interpretation. *Communication Monographs*, *85*(1), 4–40.
 https://doi.org/10.1080/03637751.2017.1352100
- Hazen, B. T., Skipper, J. B., Ezell, J. D., & Boone, C. A. (2016). Big data and predictive analytics for supply chain sustainability: A theory-driven research agenda. *Computers & Industrial Engineering*, 101, 592–598. https://doi.org/10.1016/j.cie.2016.06.030

Huber, J. (2004). New technologies and environmental innovation. Edward Elgar Publishing.

- Iivari, J., & Iivari, N. (2010). Organizational Culture and the Deployment of Agile Methods: The Competing Values Model View. In T. Dingsøyr, T. Dybå, & N. B. Moe (Eds.), Agile Software Development (pp. 203–222). Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-642-12575-1_10
- Keskin, D., Wever, R., & Brezet, H. (2020). Product innovation processes in sustainability-oriented
 ventures: A study of effectuation and causation. *Journal of Cleaner Production*, *263*, 121210.
 https://doi.org/10.1016/j.jclepro.2020.121210
- Khizar, H. M. U., Iqbal, M. J., & Rasheed, M. I. (2021). Business orientation and sustainable development: A systematic review of sustainability orientation literature and future research avenues. *Sustainable Development*, 29(5), 1001–1017. https://doi.org/10.1002/sd.2190
- Kirmse, A., Kuschicke, F., & Hoffmann, M. (2019). Industrial Big Data: From Data to Information to Actions: *Proceedings of the 4th International Conference on Internet of Things, Big Data and Security*, 137–146. https://doi.org/10.5220/0007734501370146

Kleinschmidt, E. J., De Brentani, U., & Salomo, S. (2007). Performance of Global New Product Development Programs: A Resource-Based View. *Journal of Product Innovation Management*, *24*(5), 419–441. https://doi.org/10.1111/j.1540-5885.2007.00261.x

- Kristoffersen, E., Blomsma, F., Mikalef, P., & Li, J. (2020). The smart circular economy: A digitalenabled circular strategies framework for manufacturing companies. *Journal of Business Research*, *120*, 241–261. https://doi.org/10.1016/j.jbusres.2020.07.044
- Kuckertz, A., & Wagner, M. (2010). The influence of sustainability orientation on entrepreneurial intentions—Investigating the role of business experience. *Journal of Business Venturing*, 25(5), 524–539. https://doi.org/10.1016/j.jbusvent.2009.09.001
- Kusiak, A. (2009). Innovation: A data-driven approach. *International Journal of Production Economics*, *122*(1), 440–448. https://doi.org/10.1016/j.ijpe.2009.06.025
- Kusiak, A. (2017). Smart manufacturing must embrace big data. *Nature*, *544*(7648), 23–25. https://doi.org/10.1038/544023a
- Kusiak, A., & Salustri, F. A. (2007). Computational Intelligence in Product Design Engineering: Review and Trends. IEEE Transactions on Systems, Man and Cybernetics, Part C (Applications and Reviews), 37(5), 766–778. https://doi.org/10.1109/TSMCC.2007.900669
- Laari, S., Töyli, J., Solakivi, T., & Ojala, L. (2016). Firm performance and customer-driven green supply chain management. *Journal of Cleaner Production*, *112*, 1960–1970. https://doi.org/10.1016/j.jclepro.2015.06.150
- Laskurain-Iturbe, I., Arana-Landín, G., Landeta-Manzano, B., & Uriarte-Gallastegi, N. (2021). Exploring the influence of industry 4.0 technologies on the circular economy. *Journal of Cleaner Production*, *321*, 128944. https://doi.org/10.1016/j.jclepro.2021.128944
- Lerch, C. M., Heimberger, H., Jäger, A., Horvat, D., & Schultmann, F. (2022). Al-readiness and production resilience: Empirical evidence from German manufacturing in times of the Covid-19 pandemic. *International Journal of Production Research*, 1–22. https://doi.org/10.1080/00207543.2022.2141906

- Li, C., Chen, Y., & Shang, Y. (2022). A review of industrial big data for decision making in intelligent manufacturing. *Engineering Science and Technology, an International Journal, 29*, 101021. https://doi.org/10.1016/j.jestch.2021.06.001
- Lintukangas, K., Kähkönen, A.-K., & Hallikas, J. (2019). The role of supply management innovativeness and supplier orientation in firms' sustainability performance. *Journal of Purchasing and Supply Management*, 25(4), 100558.

https://doi.org/10.1016/j.pursup.2019.100558

- Llach, J., Castro, R. D., Bikfalvi, A., & Marimon, F. (2012). The relationship between environmental management systems and organizational innovations: Environmental Management Systems and Organizational Innovations. *Human Factors and Ergonomics in Manufacturing & Service Industries*, 22(4), 307–316. https://doi.org/10.1002/hfm.20275
- Llach, J., Marquès, P., Bikfalvi, A., Simon, A., & Kraus, S. (2012). The innovativeness of family firms through the economic cycle. *Journal of Family Business Management*, *2*(2), 96–109. https://doi.org/10.1108/20436231211261853
- Manresa, A., Bikfalvi, A., & Simon, A. (2018). THE USE AND DETERMINANTS OF TRAINING AND DEVELOPMENT FOR CREATIVITY AND INNOVATION. *International Journal of Innovation Management*, 22(07), 1850062. https://doi.org/10.1142/S1363919618500627
- Manresa, A., Bikfalvi, A., & Simon, A. (2019). The impact of training and development practices on innovation and financial performance. *Industrial and Commercial Training*, *51*(7/8), 421–444. https://doi.org/10.1108/ICT-04-2019-0035

Manresa, A., Bikfalvi, A., & Simon, A. (2021). Investigating the impact of new technologies and organizational practices on operational performance: Evidence from Spanish manufacturing companies. *Central European Journal of Operations Research*, 29(4), 1317–1327.
 https://doi.org/10.1007/s10100-020-00692-8

Marco-Lajara, B., Martínez-Falcó, J., & Millán-Tudela, L. A. (2023). *Corporate sustainability as a tool for improving economic, social, and environmental performance* (p. 337). Scopus. https://doi.org/10.4018/978-1-6684-7422-8

- Marion, T. J., & Fixson, S. K. (2021). The Transformation of the Innovation Process: How Digital Tools are Changing Work, Collaboration, and Organizations in New Product Development*. *Journal* of Product Innovation Management, 38(1), 192–215. https://doi.org/10.1111/jpim.12547
- Marques, P., Bikfalvi, A., Simon, A., Llach, J., & Lerch, C. (2015). Servitisation and technological complexity in family and non-family firms: European evidence. *European J. of International Management*, *9*(2), 221. https://doi.org/10.1504/EJIM.2015.067855
- Moldavska, A., & Welo, T. (2017). The concept of sustainable manufacturing and its definitions: A content-analysis based literature review. *Journal of Cleaner Production*, *166*, 744–755. https://doi.org/10.1016/j.jclepro.2017.08.006
- Montabon, F., Sroufe, R., & Narasimhan, R. (2007). An examination of corporate reporting, environmental management practices and firm performance. *Journal of Operations Management*, 17.
- Niebel, T., Rasel, F., & Viete, S. (2019). BIG data BIG gains? Understanding the link between big data analytics and innovation. *Economics of Innovation and New Technology*, 28(3), 296–316. https://doi.org/10.1080/10438599.2018.1493075
- Nobre, G. C., & Tavares, E. (2017). Scientific literature analysis on big data and internet of things applications on circular economy: A bibliometric study. *Scientometrics*, *111*(1), 463–492. https://doi.org/10.1007/s11192-017-2281-6
- Obal, M., Morgan, T., & Joseph, G. (2020). Integrating sustainability into new product development: The role of organizational leadership and culture. 30(1), 16.
- OECD & Eurostat. (2018). Oslo Manual 2018: Guidelines for Collecting, Reporting and Using Data on Innovation, 4th Edition. OECD. https://doi.org/10.1787/9789264304604-en

- Organization, W. I. P., University, C., & INSEAD. (2018). *The Global Innovation Index 2018: Energizing the World with Innovation*. World Intellectual Property Organization.
- Palčič, I., & Prester, J. (2020). Impact of Advanced Manufacturing Technologies on Green Innovation. 14.
- Palčič, I., Pons, M., Bikfalvi, A., Llach, J., & Buchmeister, B. (2013). Analysing Energy and Material Saving Technologies' Adoption and Adopters. *Strojniški Vestnik Journal of Mechanical Engineering*, *57*(06), 409–417. https://doi.org/10.5545/sv-jme.2012.830
- Peneder, M. (2010). Technological regimes and the variety of innovation behaviour: Creating integrated taxonomies of firms and sectors. *Research Policy*, 39(3), 323–334. https://doi.org/10.1016/j.respol.2010.01.010
- Petrin, A., & Train, K. (2010). A Control Function Approach to Endogeneity in Consumer Choice Models. *Journal of Marketing Research*, *47*(1), 3–13. https://doi.org/10.1509/jmkr.47.1.3
- Podsakoff, P. M., MacKenzie, S. B., Lee, J.-Y., & Podsakoff, N. P. (2003). Common method biases in behavioral research: A critical review of the literature and recommended remedies. *Journal of Applied Psychology*, *88*(5), 879–903. https://doi.org/10.1037/0021-9010.88.5.879
- Pons, M., Bikfalvi, A., & Llach, J. (2018). Clustering product innovators: A comparison between conventional and green product innovators. *International Journal of Production Management and Engineering*, 6(1), 37. https://doi.org/10.4995/ijpme.2018.8762
- Pons, M., Bikfalvi, A., Llach, J., & Palcic, I. (2013). Exploring the impact of energy efficiency
 technologies on manufacturing firm performance. *Journal of Cleaner Production*, *52*, 134–
 144. https://doi.org/10.1016/j.jclepro.2013.03.011

Porter, M. E. (2015). Green and Competitive: Ending the Stalemate. 16.

Porter, M. E. & Claas van der Linde. (1995). *Green and Competitive: Ending the Stalemate*. 16.

Preacher, K. J., & Hayes, A. F. (2008). Asymptotic and resampling strategies for assessing and comparing indirect effects in multiple mediator models. *Behavior Research Methods*, 40(3), 879–891. https://doi.org/10.3758/BRM.40.3.879

- Pujari, D. (2006). Eco-innovation and new product development: Understanding the influences on market performance. *Technovation*, *26*(1), 76–85. https://doi.org/10.1016/j.technovation.2004.07.006
- Rehman, S. U., Bresciani, S., Yahiaoui, D., & Giacosa, E. (2022). Environmental sustainability orientation and corporate social responsibility influence on environmental performance of small and medium enterprises: The mediating effect of green capability. *Corporate Social Responsibility and Environmental Management*, csr.2293. https://doi.org/10.1002/csr.2293
- Rodríguez-Ardura, I., & Meseguer-Artola, A. (2020). Editorial: How to Prevent, Detect and Control Common Method Variance in Electronic Commerce Research. *Journal of Theoretical and Applied Electronic Commerce Research*, *15*(2), 0–0. https://doi.org/10.4067/S0718-18762020000200101
- Roxas, B., Ashill, N., & Chadee, D. (2017). Effects of Entrepreneurial and Environmental Sustainability Orientations on Firm Performance: A Study of Small Businesses in the Philippines: JOURNAL OF SMALL BUSINESS MANAGEMENT. *Journal of Small Business Management*, *55*, 163–178. https://doi.org/10.1111/jsbm.12259
- Rüßmann, M., Lorenz, M., Gerbert, P., Waldner, M., Justus, J., Engel, P., & Harnisch, M. (2015). Industry 4.0: The Future of Productivity and Growth in Manufacturing. 20.
- Saleem, H., Li, Y., Ali, Z., Ayyoub, M., Wang, Y., & Mehreen, A. (2021). Big data use and its outcomes in supply chain context: The roles of information sharing and technological innovation. *Journal of Enterprise Information Management*, 34(4), 1121–1143. https://doi.org/10.1108/JEIM-03-2020-0119
- Saleem, H., Li, Y., Ali, Z., Mehreen, A., & Mansoor, M. S. (2020). An empirical investigation on how big data analytics influence China SMEs performance: Do product and process innovation matter? *Asia Pacific Business Review*, *26*(5), 537–562. https://doi.org/10.1080/13602381.2020.1759300

Schaltegger, S., Hörisch, J., & Freeman, R. E. (2019). Business Cases for Sustainability: A Stakeholder Theory Perspective. Organization & Environment, 32(3), 191–212. https://doi.org/10.1177/1086026617722882

- Šebo, J., Šebová, M., & Palčič, I. (2021). Implementation of Circular Economy Technologies: An Empirical Study of Slovak and Slovenian Manufacturing Companies. *Sustainability*, *13*(22), 12518. https://doi.org/10.3390/su132212518
- Sen, S., & Bhattacharya, C. B. (2001). Does Doing Good Always Lead to Doing Better? Consumer Reactions to Corporate Social Responsibility. *Journal of Marketing Research*, 38(2), 225–243. https://doi.org/10.1509/jmkr.38.2.225.18838
- Serrano-García, J., Bikfalvi, A., Llach, J., & Arbeláez-Toro, J. J. (2022). Capabilities and organisational dimensions conducive to green product innovation: Evidence from Croatian and Spanish manufacturing firms. *Business Strategy and the Environment*, bse.3014. https://doi.org/10.1002/bse.3014
- Shaolin, H., Qinghua, Z., Naiquan, S., & Xiwu, L. (2021). Several Typical Paradigms of Industrial Big data Application. *Computer Science & Information Technology (CS & IT)*, 61–68. https://doi.org/10.5121/csit.2021.110906
- Shepherd, D. A., & Patzelt, H. (2011). The New Field of Sustainable Entrepreneurship: Studying Entrepreneurial Action Linking "What is to be Sustained" with "What is to be Developed". *Entrepreneurship Theory and Practice*, 35(1), 137–163. https://doi.org/10.1111/j.1540-6520.2010.00426.x
- Shou, Y., Shao, J., Lai, K., Kang, M., & Park, Y. (2019). The impact of sustainability and operations orientations on sustainable supply management and the triple bottom line. *Journal of Cleaner Production*, *240*, 118280. https://doi.org/10.1016/j.jclepro.2019.118280
- Song, M., Cen, L., Zheng, Z., Fisher, R., Liang, X., Wang, Y., & Huisingh, D. (2017). How would big data support societal development and environmental sustainability? Insights and practices. *Journal of Cleaner Production*, 142, 489–500. https://doi.org/10.1016/j.jclepro.2016.10.091

- Strange, R., & Zucchella, A. (2017). Industry 4.0, global value chains and international business. *Multinational Business Review*, 25(3), 174–184. https://doi.org/10.1108/MBR-05-2017-0028
- Surroca, J., Tribó, J. A., & Waddock, S. (2010). Corporate responsibility and financial performance: The role of intangible resources: Intangibles, Corporate Responsibility, and Financial Performance. *Strategic Management Journal*, *31*(5), 463–490. https://doi.org/10.1002/smj.820
- Survey Research Center. (2011). *Guidelines for best practice in cross-cultural surveys: Full guidelines* (3rd ed). Survey Research Center, Institute for Social Research, University of Michigan.
- Tabachnick, B. G., Fidell, L. S., & Ullman, J. B. (2007). *Using multivariate statistics* (Vol. 5). pearson Boston, MA.
- Tao, F., Qi, Q., Liu, A., & Kusiak, A. (2018). Data-driven smart manufacturing. *Journal of Manufacturing Systems*, 48, 157–169. https://doi.org/10.1016/j.jmsy.2018.01.006
- Tata, J., & Prasad, S. (2015). National cultural values, sustainability beliefs, and organizational initiatives. *Cross Cultural Management*, 22(2), 278–296. https://doi.org/10.1108/CCM-03-2014-0028
- Tehseen, S., Ramayah, T., & Sajilan, S. (2017). Testing and Controlling for Common Method Variance:
 A Review of Available Methods. *Journal of Management Sciences*, 4(2), 142–168.
 https://doi.org/10.20547/jms.2014.1704202
- Trabucchi, D., & Buganza, T. (2019). Data-driven innovation: Switching the perspective on Big Data. *European Journal of Innovation Management, 22*(1), 23–40. https://doi.org/10.1108/EJIM-01-2018-0017
- UN. (2015). TRANSFORMING OUR WORLD: THE 2030 AGENDA FOR SUSTAINABLE DEVELOPMENT -A/RES/70/1. UN.
- Usman Khizar, H. M., Iqbal, M. J., Khalid, J., & Adomako, S. (2022). Addressing the conceptualization and measurement challenges of sustainability orientation: A systematic review and research

agenda. Journal of Business Research, 142, 718–743.

https://doi.org/10.1016/j.jbusres.2022.01.029

- Vilkas, M., Bikfalvi, A., Rauleckas, R., & Marcinkevicius, G. (2022). The interplay between product innovation and servitization: The mediating role of digitalization. *Journal of Business & Industrial Marketing*, *37*(11), 2169–2184. https://doi.org/10.1108/JBIM-03-2021-0182
- Wamba, S. F., Akter, S., Edwards, A., Chopin, G., & Gnanzou, D. (2015). How 'big data'can make big impact: Findings from a systematic review and a longitudinal case study. *International Journal of Production Economics*, *165*, 234–246.
- Wang, J., & Dai, J. (2018). Sustainable supply chain management practices and performance. *Industrial Management & Data Systems*, 118(1), 2–21. https://doi.org/10.1108/IMDS-12-2016-0540
- Ward, J. S., & Barker, A. (2013). Undefined By Data: A Survey of Big Data Definitions. *arXiv:1309.5821* [Cs]. http://arxiv.org/abs/1309.5821
- Yudhistyra, W. I., Risal, E. M., Raungratanaamporn, I., & Ratanavaraha, V. (2020a). Exploring Big Data Research: A Review of Published Articles from 2010 to 2018 Related to Logistics and Supply Chains. *Operations and Supply Chain Management: An International Journal*, 134–149. https://doi.org/10.31387/oscm0410258
- Yudhistyra, W. I., Risal, E. M., Raungratanaamporn, I., & Ratanavaraha, V. (2020b). Using Big Data Analytics for Decision Making: Analyzing Customer Behavior using Association Rule Mining in a Gold, Silver, and Precious Metal Trading Company in Indonesia. *International Journal of Data Science*, 1(2), 57–71. https://doi.org/10.18517/ijods.1.2.57-71.2020
- Zhang, X., Ming, X., & Yin, D. (2020). Application of industrial big data for smart manufacturing in product service system based on system engineering using fuzzy DEMATEL. *Journal of Cleaner Production*, *265*, 121863. https://doi.org/10.1016/j.jclepro.2020.121863
- Zhao, M., Yang, J., Shu, C., & Liu, J. (2021). Sustainability orientation, the adoption of 3D printing technologies, and new product performance: A cross-institutional study of American and

Indian firms. *Technovation*, 101, 102197.

https://doi.org/10.1016/j.technovation.2020.102197

- Zhong, R. Y., Newman, S. T., Huang, G. Q., & Lan, S. (2016). Big Data for supply chain management in the service and manufacturing sectors: Challenges, opportunities, and future perspectives. *Computers & Industrial Engineering*, *101*, 572–591. https://doi.org/10.1016/j.cie.2016.07.013
- Zhou, Z.-H., Chawla, N. V., Jin, Y., & Williams, G. J. (2014). Big data opportunities and challenges:
 Discussions from data analytics perspectives [Discussion Forum]. *IEEE Computational Intelligence Magazine*, 9(4), 62–74. Scopus. https://doi.org/10.1109/MCI.2014.2350953